Content and Structure of Laboratory Packages for Software Engineering Experiments

Martín Solari^{a,*}

Sira Vegas ^b

Natalia Juristo ^b

^a Universidad ORT Uruguay, Cuareim 1451, 11100 Montevideo, Uruguay. ^b Universidad Politécnica de Madrid, Campus de Montegancedo s/n, 28660 Madrid, Spain. Email: <u>martin.solari@ort.edu.uy</u> (M. Solari), <u>svegas@fi.upm.es</u> (S. Vegas), <u>natalia@fi.upm.es</u> (N. Juristo) * corresponding author

Context: Experiment replications play a central role in the scientific method. Although software engineering experimentation has matured a great deal, the number of experiment replications is still relatively small. Software engineering experiments are composed of complex concepts, procedures and artefacts. Laboratory packages are a means of transferring knowledge among researchers to facilitate experiment replications.

Objective: This paper investigates the experiment replication process to find out what information is needed to successfully replicate an experiment. Our objective is to propose the content and structure of laboratory packages for software engineering experiments.

Method: We evaluated seven replications of three different families of experiments. Each replication had a different experimenter who was, at the time, unfamiliar with the experiment. During the first iterations of the study, we identified experimental incidents and then proposed a laboratory package structure that addressed these incidents, including document usability improvements. We used the later iterations to validate and generalize the laboratory package structure for use in all software engineering experiments. We aimed to solve a specific problem, while at the same time looking at how to contribute to the body of knowledge on laboratory packages.

Results: We generated a laboratory package for three different experiments. These packages eased the replication of the respective experiments. The evaluation that we conducted shows that the laboratory package proposal is acceptable and reduces the effort currently required to replicate experiments in software engineering.

Conclusion: We think that the content and structure that we propose for laboratory packages can be useful for other software engineering experiments.

Keywords: software engineering, experiment replication, laboratory package, knowledge transfer.

1 Introduction

Experimentation is a way of maturing knowledge. It is at the heart of the scientific method. Experiments manipulate and observe reality in a controlled, rigorous and systematic manner in order to discover variables that somehow influence a specified phenomenon [1]. For this approach to be successful, experimentation must be iterative, taking earlier results to refine procedures for studying the phenomenon in more depth. As a research method in software engineering (SE), experimentation dates back only a few decades having first been formalized around 1986 [2].

Replication is a way of exploring experiments in more breadth and depth [3]. Replications serve two purposes: (i) they may study whether the results are valid for other contexts or whether other researchers are able to get the same results, and (ii) they may, if changes are made to an experiment, allow studying influencing variables and provide information about the bounds of the acquired knowledge [4]. Since replication is troublesome in SE, some researchers [5] have proposed alternative methods. Although there are many methodological uncertainties surrounding replication, its importance is clear [6].

Experiments are composed of concepts and of particular artefacts for experiment operation. The conceptual elements include the constructs, hypotheses and experimental design. The operational elements include the instruments, materials and procedures. On this ground, an experiment is a highly complex entity both conceptually and operationally. This is an obstacle to its replication, particularly when a researcher is replicating an experiment that he or she did not design [7].

There are some ways of facilitating experiment replication. They include collaboration among researchers and laboratory packages [8] [9] [10]. The use of such types of knowledge transfer mechanisms is widespread among researchers from other disciplines, such as physics, biology and the social sciences [11] [12] [13].

Laboratory packages (LP) contain the information and materials required to replicate an experiment. The term protocol is often used in natural science [14] to refer to the packaging of key information about an experiment for its transmission and use in a replication by others. Whereas the term protocol underscores strict adherence to a procedure, the term LP focuses on the transfer of information in self-contained units. Both components are necessary for conducting a replication.

The content of a LP is not static; it needs to be adapted to the needs of the researcher running the replication. A LP is usually task driven (that is, written like a *recipe*) and may use alternatives other than text like video or software. A scientific paper is unlikely to include this type of content as there are usually constraints on space and format [15]. Scientific publications usually stress the high-level conceptual aspects, omitting more specific operational issues that are helpful for replicating experiments. In some disciplines, this gap has been filled by journals dealing specifically with experimental protocols like *Nature Protocols*¹. Open digital repositories of experimental protocols have also been set

¹ http://www.nature.com/nprot/index.html

up, providing for discussion among researchers based on a forum system². These repositories enable a faster information exchange than journals, as well as the use of alternative formats for answering questions and helping other researchers to replicate the experiment.

Although SE has made significant progress with regard to the conceptualization and running of replications, information packaging and transfer are still an open question. The use of LPs has been defended ever since the earliest systematic experiences of SE experiment replication [7]. The need for LPs as a key component for conducting families of experiments was again stated later on [8]. Collaboration between groups of SE experimenters was the focus of research project that proposed the use of a knowledge transfer model also based on LPs [16]. Interest in the topic appears to have cooled within the SE community, but this does not mean that how to package and transfer the information needed to run a replication has been solved. Although the number of replications has grown over the last few years, it is still not a common practice, even less so if researchers are trying repeat an experiment at other sites [17]. Neither are there many cases of SE experiments whose materials are published for use by other researchers, and experiments whose procedures are shared with a view to conducting exact replications are even rarer still [18].

We aim to identify which information a LP for SE experiments should contain. Proper LPs would encourage more and better replications. Replications not only contribute to the body of SE knowledge, but also help researchers to learn about how to experiment with and observe the phenomenon. This helps to mature SE experimentation.

Our immediate goal is to improve the LPs used to replicate experiments within a SE experiment network. Although the research was undertaken in a small experimental research network, we believe this context is representative of the current replication and information exchange scenario between experimental researchers in SE.

We studied three experiments and seven replications iteratively. First, we observed the incidents in replications of one experiment run by researchers belonging to several research groups. From these observations we derived a preliminary LP structure. In later iterations, we refined the structure by observing more replications of the same experiment run by researchers belonging to the same and other research groups. Finally, we generalized the structure to build LPs for other two experiments.

The structure of the paper is as follows. Section 2 discusses related work, analysing the use of LPs and experimental protocols first in other disciplines and then in SE. Section 3 describes the research method. The other sections each address one of the stages of the research. Section 4 reports the diagnosis based on the observed incidents in experiment replications. Section 5 outlines the proposed LP structure. Section 6 describes the LPs generation for three experiments. Section 7 describes how LPs were evaluated by observing replications using them. Section 8 reflects on how the proposed LP might be generalized. Section 9 present further evolution plans for our proposal. Finally, Section 10 outlines the conclusions of the research.

² http://www.nature.com/protocolexchange/

2 Related Work

2.1 LPs in other disciplines

It is common practice in many scientific disciplines to use LPs in different types of experimental replications. In the social sciences, for example, it is usual to share sampling, observation or intervention procedures for complex phenomena [13]. Disciplines like biology and medicine use repositories containing the protocols to be followed by experimenters [19]. These protocols facilitate the aggregation of studies replicated by different researchers at different sites.

Since the 1990s several scientific communities have helped to create new methods for speeding up research and promoting information exchange [20]. Digital media are not subject to the typical constraints on number of pages and format imposed by the print media. They also facilitate discussion and contacts among researchers.

Giles [15] argued in an opinion column that researchers visit each other to learn the protocol because much of the knowledge about the experiment is tacit. The column cites several scientific publication editors who stress the need to publish more detailed information and provide for direct exchange among the researchers that are going to run replications. Examples of journals that provide these exchange mechanisms are: *PLoS ONE*³, *Cold Spring Harbor Protocols*⁴ and *Nature Protocols*⁵.

Protocol publication and study preregistration have become the norm in medicine. In recognition of the importance of detail in order to ensure the rigour of clinical trials, several initiatives try to standardize the content of such reports and protocols. The CON-SORT (CONsolidated Standards of Reporting Trials) initiative promotes study reporting quality by means of a content checklist that was originally proposed in 1997 and has been updated on several occasions [21].

There is agreement among the editors of several medical journals about the importance of having access to the full protocol of published trials [14] [22]. Journals like the *Lancet⁶*, *British Medical Journal*⁷ and *PLoS Medicine*⁸ demand that the protocols be submitted together with controlled experiment reports for inclusion in the review process. However, current guidelines on protocol content vary substantially with respect to recommendations [23]. Recently, the SPIRIT (Standard Protocol Items: Recommendations for Interventional Trials) initiative proposed the standardization of the protocol content of experiments in medicine [24].

³ http://www.plosone.org/

⁴ http://cshprotocols.cshlp.org/

⁵ http://www.nature.com/nprot/index.html

⁶ http://www.thelancet.com/protocol-reviews

⁷ http://www.bmj.com/about-bmj/resources-authors

⁸ http://www.plosmedicine.org/static/guidelines

An exceptional example of LP or protocol repository is *Nature Protocol Exchange*⁹. This repository provides for the open exchange of experimental protocols and an open discussion. Replicating experimenters can submit queries and baseline experiment authors can provide the details that they had omitted to specify. Unlike *Nature Protocols*, the repository does not follow a process of arbitration before the protocols are uploaded. Research groups can sign up to share their own protocols or query the protocols of other groups. The repository is open to experiments from any scientific discipline. These procedures and data can be taken by other researchers to run replications or to perform secondary studies (systematic literature review (SLR), meta-analysis, etc.).

Although there is no mandatory structure, the suggested format for uploading protocols is the same as for submissions to *Nature Protocols*. This protocol format is composed of the standard sections of a scientific paper (authors, abstract, introduction, etc.), but also by sections specializing in experiment replication like:

- Procedure
- Timing
- Troubleshooting
- Anticipated Results
- Supplementary Information

Step-by-step guidance should be provided for the Procedure section. The Supplementary Information section may include almost any type of digital information like audio, video, equations, methods, figures, etc. This type of content clearly goes beyond the dissemination of the findings, aiming to record detailed procedures and materials for other experimenters to be able to replicate the experiment.

The *Nature Protocol Exchange* supports a more fluent discussion between researchers conducting the same experiment because it accommodates different types of materials, enabling and encouraging the inclusion of more detailed information about procedures and provides a rapid exchange process. Note that all these initiatives share the following implications:

- Detailed information should be supplied for an experiment to be able to be replicated,
- Information about an experiment is continuously evolving [15].

A LP follows the dynamics of the experimental process. Rather than a closed file, it is an open information exchange tool within a research process. Replications are a means for both building confidence in the results and learning new variables about the phenomenon under study [25]. Replications are also a way to learn about and refine the methods of observation, that is, learn more about the experiment itself. On this ground, the interaction between LPs and the researchers that use them is complex, and they are an intrinsic and evolving part of experimental research.

⁹ http://www.nature.com/protocolexchange/

2.2 LPs in SE

The LP issue has been addressed on several occasions by the SE community. The community recognized the need for LPs to promote more replications in the 1990s [7] [26]. In the late 1990s, Basili proposed the concept of families of experiments as a way of achieving coordinated replications based on a LP [8].

Shull revived the idea of transferring knowledge among researchers when he published one of the most sophisticated examples of LP. This LP, published in web format, accounts for multiple replications run within the same framework. Apart from the LP itself, Shull et al. propose an explicit knowledge management model to facilitate information exchange within the replication process [27].

After the publication of this LP, the SE community's interest in this topic waned. Although these early papers succeeded in raising the awareness of many experimenters about the need to publish the experimental materials for the purposes of replication, the experiments run do not tend in practice to provide LPs that are as mature as the one proposed by Shull. Many published experiments are supplemented by information available on the web, but hardly any include anything more than the experimental material (for example, programs, data collection sheets) or raw data supporting the results [18]. Neither of these options is sufficient for running a replication.

Proposals for standardizing controlled experiment reporting were published in the early 21st century [28]. They were well received by the SE community. This proposed reporting format contains specific sections constituting the groundwork for a publishable paper. However, the focus on structuring the publication of experiment results omits the materials and procedures required for replication. A preliminary proposal specifically for reporting experiment replications was made later [29], but again it does not include a structure for publishing materials and procedures to enable replication to continue.

Recently a few groups have systematized the publication of materials and procedures for their experiments, making the information available on the web. Two notable examples are the Software Engineering Research Group's (SERG)¹⁰ site at Lund University, which contains LPs for more than 20 experiments, and Lutz Prechelt's page at the Free University of Berlin, which contains a section with materials and data for replicating eight experiments¹¹. The structure of the LPs published at these sites is ad hoc and depends on the experiment. LPs are not standardized even within the same group. Both groups have encouraged the replication of their experiments and put the accent on information sharing with other researchers.

The systematic mapping study conducted by Da Silva reported 133 replications of SE experiments. They found that the number of replications is growing, and there is more and more discussion about the importance of having a collaborative agenda among researchers [17]. However, the mapping does not identify any theoretical inputs for building LPs, except for the ones mentioned at the beginning of this section. They state that many

¹⁰ http://serg.cs.lth.se/index.php?id=32563

¹¹ http://page.mi.fu-berlin.de/prechelt/Biblio/#package

examples of LPs have been published, but problems such as the proper recognition of the contributions of different researchers persist [30].

In order to form a closer idea of the state of LP practice with regard to SE experiments, we have extended the review that we published back in 2006 [18]. We searched the public web (using the Google search engine) instead of confining the results to scientific bibliographic databases. The keywords used for the search were: *software engineering replication* (alternative *laboratory*) *package*. We manually reviewed the primary results to see which really did contain LPs. We also added LPs with which we were personally acquainted about to the review manually. We only considered LPs that were publicly available online (which is the only way of guaranteeing access for potential replicating experimenters). Some of the LPs were not available at the time that we concluded this research. However, they are considered because they were available when we conducted the original search in 2006 [18], updated in 2010 [31] and again in 2014 for this paper.

We divide the identified SE experiment LPs into three categories: single experiments, material repositories and general-purpose support infrastructures:

- LPs built for a **single experiment** or single family of experiments usually have a specialized structure suitable for describing the particular characteristics of the experiment. It is common practice for this type of LP to be referenced from scientific papers that, for reasons of space, cannot include all the details of the experiment. An example of this type of LP is the UML comprehension experiment published by Genero et al. [32].
- **Material repositories** are a special case of LP focusing on experimental objects. For example, they contain different versions of programs and their respective information. These objects can be used in different types of experiments. An example of this type of LP is the repository of artefacts for testing experimentation created by Do et al. [33].
- **General-purpose support infrastructures** are not confined to a specific family of experiments. On the contrary, they can support the activities of the experimental process for any family of experiments. The structures may be conceptual (for example, data models, ontologies) or be implemented in a specified system. We have only considered support repositories or infrastructures that have at least one instance of a specific experiment. An example of this type of LP is the computational framework proposed by Mian et al. [34].

Our research will focus on the first type of LP. We believe that a LP should be a selfcontained tool that is independent of other documents. On this ground, we examine the information that the LP contains rather than information that is scattered across articles published about different experiments. This does not mean that other types of LPs are not needed in order to take replication forward, but merely that they are beyond the scope of our research at this stage.

Table 1 gives an overview of the LPs identified for single experiments. We found that the number of publicly available LPs is rather low compared with the number of experiments

and replications in SE [35] [17]. Whereas material or additional descriptions may have been published in some medium or another, they are not easy for web search engines to locate. The repositories of SERG at Lund University and Prechelt's page at Fee University of Berlin were not found in our initial search and comparison. Both repositories have LPs for multiple experiments and significantly improve the state of published material in SE. However, the repositories don't use uniform LP structures.

LP	1	2	3	4	5	6	7
Authors	Kamsties & Lott [36]	Basili et al. [37] Shull et al. [16]	Thelin et al. [38]	Dunsmore [39]	Do et al. [33]	Du Bois [40]	Genero et al. [32]
Year	1995	1995 2004	2003	2003	2005	2006	2008
Research topic	Functional, and struc- tural testing, code review	Perspec- tive-based reading	Checklist- based reading	Object- oriented inspection	Regression testing	Class decompo- sition	UML stereo- types
Known replications (institution)	Kaiserslau- tern Strathclyde	Maryland NASA SEL Kaisers- lautern	Lund	Strath- clyde	Nebraska	Antwerp	Bari Castile-La Mancha

Table 1: Overview of identified LPs.

Most of the identified LPs are appendices to published papers. On one hand, scientific papers must adhere to journal or conference formats and space constraints and often do not include specific replication guidelines.

We found that the LPs also include scant descriptions. In most of the cases, these descriptions do not target the replicating experimenter but are designed to provide materials or data to supplement a paper. While this is of no assistance for encouraging replication, it does help to make experimental results more transparent. The sections of the analysed LPs differ substantially and do not account for different experimental process activities in the same manner. We have also found differences in the language used to describe the same experimental concepts.

An all-round analysis of the identified LPs gives an idea of the key weaknesses of SE LPs:

- Most LPs focus on the artefacts for operating the experiments. Some do not contain any more information than the operational material. Generally, this is because they assume that experimenters will have read a paper before consulting the LP. In these cases, we suggest that the term material be used to make it clear that it has a different objective than LPs. In fact, such material might be referred to as supplementary information to make its purpose clear.
- Descriptions and guidelines for replication activities are less common. Instructions targeting replicating experimenters have only ever once been built into the LP format itself [16].

- No LPs have included information on replications conducted in the past or the evolution of the experimental research (extracted pieces of knowledge and their interpretation), save on the occasion of one experiment [16]. The replication findings are usually published as part of papers or technical reports but are not incorporated with all the other information into one and the same structure.
- LPs do not have a standard structure, and different formats are used to organize the content. Although most of the studied LPs are based on a web page and links to files in other formats, there are a vast number of formats and configurations.

Therefore, the problem of how to package the information required to replicate an experiment in SE can by no means be considered to have been solved. Yet we build our solution on some of the previous research.

The starting point of our proposal is the analysis of previously published LPs for SE experiments. Most of these LPs include the experiment materials as an appendix to a published article. We found that scientific journals had space and format limitations, and the materials did not include any replication guidelines. Therefore, we wanted to address the problem using a self-contained structure.

One notable exception is Shull et al.'s LP [16], which is specifically constructed to help different groups replicate an experiment. This LP provides a description of the experiment as separate from the report for each replication. It also includes a history section reporting how the experiment evolved. This LP had a considerable influence in our proposal. The structure that we propose reuses three key components already present in Shull et al.'s LP:

- includes specific guidelines for replicating the experiment,
- separates the description of the baseline experiment from the report for each replication,
- summarizes the evolution of the experiment inside the LP structure.

3 Research method

The need of LP emerged as a necessity when internally and externally replicating an experiment. Over 14 years ago, the experimental research group at the Technical University of Madrid (UPM)¹² contributed to a family of testing experiments [41]. We received the materials from one of their replications [42] which was based on a previously published LP [36]. In the first place, we had to study and organize these materials to produce an experimental design adapted to the particular conditions of the UPM replication. This experiment was internally replicated three times by the same researchers [43]. Later, this experiment was again internally replicated (at UPM) but by other researchers from the same group. This meant that a LP had to be put together to transfer the information. At a later stage, the experiment was replicated under an arrangement with other groups and again the LP had to be adapted in order to conduct external replications [9] at Universidad de Sevilla (UdS), Universitat Politècnica de València (UPV) and Universidad ORT Uruguay

¹² http://www.grise.upm.es/

(ORT). The experiment was modified according to the recorded outcomes, and the LP had to include this new information [44]. Other groups are still replicating the experiment using the LP [45] [46] [47].

This research commenced in 2006 with the definition of a framework for observing experimental replications and evaluating LPs [48]. We applied this framework in each iteration of the research. Thanks to the evaluation framework, we were able to identify any incidents occurring that were an obstacle to or jeopardized the results of the replication. We thoroughly examined each particular incident that occurred, relating the incidents across different replications. The incidents were grouped for later generalization in order to identify archetypal incidents likely to occur when replicating experiments in SE.

The major sources of information were the researchers participating in the replication of the experiment and the LPs used in each replication (including documents and any form of communication). The researchers were interviewed after a replication. The strategy underlying the research iterations is to gradually alter experiments and replication sites. Table 2 shows the iterations in the research method and the different experiments and research groups involved in each iteration.

	Iteration 1	Iteration 2	Iteration 3	Iteration 4
Experiment	DefDetect	DefDetect	DefDetect	UnitTest and TeamPersInf
Replications	UPM 2000	UPM 2004, UdS 2005, ORT 2005, UPV 2006	ORT 2009, UPM 2011	
Diagnosis	Experimental incident analysis	Experimental incident analysis	Experimental incident analysis	
Intervention		Draft of proposed LP to help mitigate identified incidents	LP instantiation	LP instantiation
Evaluation			Evaluation of LP use in experiment replications	Evaluation of the instantiation of the proposed LP

|--|

In Iteration 1, we ran a diagnosis of the replications of an experiment about defect detection techniques (DefDetect experiment) run within our research group [49]. We had replicated the experiment using a LP provided by another group [42]. After analysing what information we would have liked to have had about the experiment for replication, we generated a new LP for this same experiment including this information for use in successive replications.

In Iteration 2, we extended the diagnosis to replications of the same experiment conducted at other sites. The external replications [9] used an improved version of the LP. We devised a LP proposal that met the needs revealed by these three new replications. We used the identified experimental incidents to propose the structure of a LP to mitigate such incidents in replications generating a new LP version for the DefDetect experiment. In iteration 3, we evaluated the experimental incidents in replications using the new LP. Once the structure and the LP structure had been stabilized for the DefDetect experiment, we moved on to other experiments with the aim of generalizing the proposed LP structure. In Iteration 4, we validated and refined our proposed LP, generating LPs for another two experiments run and replicated by other research groups. These experiments and research groups were progressively further distant from the original experiment and group. In particular, we generated LPs (based on the structure proposed in Iteration 2) for another two experiments: another experiment by the UPM group (conducted by different researchers than the DefDetect experiments) on team and personality influence in development (TeamPersInf experiment) replicated at [50] and an experiment on unit testing techniques (UnitTest experiment) [51] by a research group that is not a member of the original network (University of the Republic, Uruguay).

4 Analysis of experimental incidents

In the diagnosis stage, we analysed experimental incidents in several replications. The analysis focused on the problems and obstacles occurring in each individual replication. Researchers running replications all have different backgrounds in terms of both experimentation and the experiment domain. Therefore, they have different information needs when it comes to replicating an experiment. On this ground, their subjective views of the process of replication have to be taken into account. As the phenomenon in question is complex, we used a qualitative evaluation method. Evaluation is empirical as we observe the use of the LP as part of several separate replications (one at a time).

During the evaluation we analysed the instruments and, particularly, the LP used during the replication. We identified and studied particular experimental incidents and tried to determine the extent to which the LP influences each incident [48]. We classified the incidents by types and identified the needs that the LP was expected to satisfy. The specific goals of the evaluation of the replications were:

- 1) Discover the purpose and setting in which the replication was run,
- 2) Discover the characteristics of the LP used, as well as other instruments of communication used to run the replication,
- 3) Identify the incidents in each experimental activity,
- 4) Understand the viewpoint of the replicating experimenter with respect to the LP and to the development of the replication.

In order not to interfere in the replication, the evaluation was conducted post-mortem at the end of each replication. As replication is a highly intellectual activity, mere observation (without interaction with experimenters) did not show up enough information in order to understand what was going on. The key information-gathering activity was a semistructured face-to-face interview [52] with the experimenter responsible for each replication. Thanks to this interview method, we were able to gather a considerable amount of information from the key source and rapidly answer any questions. Before the interview, all available documents associated with the replication, including the replication LP itself, were collected. This preliminary phase improved the effectiveness of the interviews as it put things into context. Also, experimenters were contacted by email if any further questions arose at a later date. The full responses were used to build a systematic transcript and then a structured summary for each replication. The summary uses coded fields to describe each replication using a metalanguage that evens out the differences between the literal responses of each experimenter.

The coding was applied iteratively to output a list of experimental incidents. The codes of each incident were not generated a priori; they were output iteratively based on observations made during the evaluation of the replications. First, one researcher encoded the incidents of each particular replication and then the same researcher analysed the similarities with the other evaluated replications. The same code was used to denote identified incidents that are comparable to each other. Another researcher reviewed the mappings between the generated codes and the systematic transcripts. As a result, instead of the literal expressions used by the replicating experimenter during the interview, the detected incidents are designated using a standardized code that stands for related problems and obstacles.

We grouped the incidents by categories in accordance with the experimental process activities. Table 3 summarizes the identified incidents. Table 3 also shows the output of the diagnosis stage for the first two iterations. For a detailed description of each incident and the process enacted for their identification, see [48].

The LP and inter-researcher communication mechanisms used in each replication were different even in replications of the same experiment conducted close together in time. The standard LP for the networked replications from 2005 to 2006 consisted of a main document and related files. But this document evolved over time, and each version omitted some information. Some elements were communicated by email or at meetings depending on the replication. These variations led to specific incidents, such as the failure to include defect information for programs or some tasks requiring more effort. Therefore, the definition of incident is dependent on the context of each replication. What is considered to be an incident in some scenarios will not necessarily be an incident in others.

The identified incidents are always based on the viewpoint of the researcher who is replicating the experiment in a specific context. This means that experimenters with different backgrounds may identify different incidents in some contexts. For example, irrespective of the information included in the LP, a researcher could be unfamiliar with the experiment objects. For instance, none of the LP versions contained guidelines for data analysis. In a replication where researchers are already familiar with data analysis techniques or another group performed the data analysis, this may be not perceived as an incident, since the incident is defined from the researcher's perspective.

Activity	Incidents	Itera	tion 1	1	teration	2
		UPM	UPM	UdS	ORT	UPV
		2000	2004	2005	2005	2006
Communication	No communication	X				
	Communication without direct contact			Х		
	Limited or after-the-event communication				Х	
	No meeting held to validate design			Х	X	
	No in-person observation of sessions			X	X	
	Renlicator felt cut off			X		
	There were a lot of misaivinas	х			X	
	Experiment is hard to understand		x			
Exportmontal	Loss time assigned to one of the treatments		А			v
design	Easton grossing owner			v		Λ V
uesiyn	Non randomized assignment of subgroups				v	
	Inhalanced experimental around			A V	Λ	Λ
	Missivings shout the impact of the design shouse	v		Λ	v	
	Misgivings about the impact of the design change	X	1		X	
T	Diderestimated experiment adaptation workload	X		v	X	v
Iraining		V		Ă	X	X
	Other material used in training	X			X	X
	Training received did not match the treatment			X		X
	A treatment (functional technique) was misunderstood			Х		
	A treatment (structural technique) was misunderstood				Х	Х
Material	Too much material for one treatment					Х
preparation	The supplementary sheet was removed			Х		
	Line numbers were added to the source code				Х	Х
	One of the programs could not be compiled on the platform	Х				
	Support from other people was required to prepare the		Х			Х
	Inderestimated material propagation workload			v	v	
	Four that material contingencies would affect session			Λ	л	
	time				Х	Х
	Compley time concurring activity	v		v	v	v
Operation	Complex, time-consuming activity	Λ		A V	A V	Λ V
operation	Participants ask a lot of questions			Λ	Λ	A V
	Participants do not redd dil the material					Λ
	Researcher is unfamiliar with objects and cannot		Х			Х
	Limited accessor time			v	v	
	Limited session time			Λ	A V	
	Rigorous atmosphere of session				X	
	Effect of fatigue on participants	V			X	
D (1 (Some participants apply wrong treatment	X		17		
Data analysis	No analysis was conducted			X		
	Analysis was postponed				X	
	Support from another person was required to complete analysis					Х
	Correctness criterion defined by experimenter			Х	Х	Х
	Test cases and faults written by participants are					
	ambiguous	Х	Х			Х
	Correction was perceived as very complicated		Х			Х
	No guidelines or examples were available for the					
	analysis					Х
	No previous data were available to compare results	Х	Ì			
	Fault description was not integrated into the laborato-					
	ry package					Х
	Fault description was not detailed enough	Х	t	1	1	1
	Errors in the fault/failure description			Х		
Research	Notes taken on the replication were destroyed		x		x	x
process	Replication reporting was postnoned	x		x	X	
r	Research cycle was incomplete		1	X	X	x
1	Experiment was not evolved		1	X	X	x
			1			

Table 3: Incidents identified across replications.

5 Designing a LP proposal

The LP proposal is composed of a set of elements aiming to improve the information available to the replicating experimenter and avoiding incidents in the replication. The proposed LP supports *all* the activities of the experimentation process and not only execution. In order to identify the activities of the experimental process, we used the proposals in different experimentation books [53] [1]. In order to provide a complete description of the experiment, the design of our LP proposal considers Jedlitschka's guidelines for reporting experiments [28] as well as Carver's specific guidelines for reporting replications [29]. We also considered facets of the technical documentation quality [54] [55] [56]. These facets include overall usability of the document containing the LP, information navigability, search and retrievability. These cross-cutting issues are evaluated in each version of the instantiated LP. The complete description of the proposed LP is available as a web appendix in our website¹³.

For the initial design of the LP proposal, we used the experimental incident list as main guidance for generating specific components in the structure. We analysed experimental incidents by category and hypothesized which components could have prevented each incident. After generating an initial design, we mapped each one of the identified incidents to one or more components of the proposed LP in a comprehensive relationship matrix. This matrix covered all identified experimental incidents to assure that each one was addressed by at least one component of the proposal. This relationship matrix is included in the web appendix.

5.1 Proposed LP structure

Our LP proposal is divided into modules. We think experimenters might have very diverse information needs. Each experimenter has a different background and distinct experience in experimentation. We have found that an experiment is usually run by several researchers who play different roles. The structure of a LP should enable each researcher to easily identify the key information for the task that he or she is performing without having to leaf through pages of information that have nothing to do with the responsibilities of his or her role. The aim of dividing the LP into modules is to cover all the phases of the experimental process and meet the information needs of different researchers depending on what activity of the experimentation process they are to perform in a manner that is clear to all LP users.

The modules constitute separate parts of the document. However, they are explicitly linked to each other so that anyone who wants to read all the documentation instead of focusing on a particular activity will find it easy to navigate from one to another. We see a module as a structure that is used separately for some experimental activity. For example, the replicating experimenters use one module to train subjects, another to adapt the design and another to record the results of the replication. We have tried to make each module as self-contained as possible in order to satisfy information needs without the researcher having to resort to other parts of the documentation. If necessary, the modules are further divided in order to include more specific information on diverse subtasks. For

¹³ http://www.grise.upm.es/LaboratoryPackage

example, the experiment module is subdivided into sections on experimental design, session operation and data analysis.

We expect that running a replication should also imply upgrading the LP to integrate the new piece of generated evidence that will help to mature the knowledge pursued by the family of experiments (and supported by the LP). After a replication, it should at least be added to the experiment log generating a new module containing the detailed description of the replication and its results. As a result of the replication, however, observations are often made and lessons learned about the experiment itself, the support materials or even the theory. Therefore, a replication may generate a new version of the LP modules containing this information. Note that the generation of new versions of modules should never result in the replacement of earlier versions, that is, the package also acts as a historical record.

Figure 1 shows the main structure of the LP with the name of each module. The directed lines denote the dependencies between modules. This entire structure is the starting point. Elements can then be added to the LP to mitigate the experimental incidents that we have identified. We have tried to design a preliminary structure that is as detailed as possible to cover the information needs and experimentation process activities. Each module is subdivided into sections. These sections provide help for mitigating the missing information detected in the incidents of a particular experimental subtask.





The **introduction to the LP** module contains information about the LP itself. This module is at a different level of abstraction to the other modules. Whereas the other modules refer to the experiment, this module refers to the actual LP. The Introduction module is designed to provide the LP user with information about the structure and use of the LP. The LP users may be researchers intending to replicate an experiment or researchers setting out to package an experiment for replication by others. The Introduction module denotes the aims of the instrument (LP), possible uses and how it can be enriched. It also includes the suggested communication instances and the particulars of the researchers who should be contacted in order to run a replication.

There are two other types of LP modules: **core modules** and **study modules**. The core modules are modules that contain basic information about the experiment and its underlying theory. While there is only one instance of these modules in the LP, there should be more than one version for the purposes of LP configuration management. The core modules must be modified when significant changes are made to the experiments, thus generating new versions. The study modules each match a replication or aggregation conducted as part of the family of experiments. The number of instances of the study modules in the LP is equal to the number of studies that have been conducted. The study modules are added to the structure of the LP with each replication or aggregation, without altering the previous study modules.

The **core modules** have a higher level of abstraction and contain the groundwork of an experiment. These modules include the theory, the instructions for replicating the experiment and training material. The module with the highest level of abstraction is the experiment **theory**. It establishes the theoretical framework for conducting the experiment and interpreting the results. All the other modules are directly or indirectly related to the theory module. The **experiment** module defines the experiment to be replicated. This module includes the instructions for the replicating experimenter, the operational material and the experimental tools. The **training** module contains all the materials used to train the experimental subjects (if applicable).

The **study modules** are added to the LP as the replications or aggregations of the experiment are conducted. Instances of these modules can be added for every replication or secondary study carried out with LP data. The **replication** modules capture information about each replication. The replicating experimenter is expected to add a module of this type describing his or her replication to the LP. Each module includes the description of the adapted design, the resulting data and notes on the replication experience. The **aggregation** modules are for recording the secondary analyses conducted on the replication data. These modules record anything from the inspection protocols, meta-analyses or other comparison techniques used and the findings of the secondary study.

The **evolution** module is a special case within the LP. There is only one instance of this module, and it is therefore an experiment module. However, a new section is aggregated to the evolution for each and every one of the replications and aggregations added to the LP. This module serves the purpose of establishing a relationship between the experiment versions and their replications, as well as of picturing how the LP evolves as a whole. For example, when several replications are conducted by different research groups, the exper-

iment often tends to evolve and the experimental objective to change (in order to further generalize the results). This module should reflect the historical evolution of the experiment and the replications to which each version of the experiment is linked.

When it is necessary to modify any part of the experiment, the changes should be made within one and the same module whenever possible. However, there are often external dependencies. In order to make the LP easier to read, module dependencies are one-way only. The lower-level modules depend on the higher-level modules. For example, the replication modules depend on the experiment module, and this depends on the theory module. Generally, this avoids circular references and eases LP evolution.

5.2 LP template

Each module of the proposed LP structure is further divided into sections and subsections. These sections are described in detail in a LP template document. The LP template is a tool for the researcher who is responsible for instantiating an experiment. The LP template assigns a correlative number to and provides a content description for each section and subsection. Table 4 shows an excerpt of the LP template for the experiment module. It gives a sample content description for one subsection in each section of the module, omitting the other subsections for brevity. The LP template document with the content description for each section is included in the web appendix.

Section	Subsection	Content
4.1 Plannina	4.1.1 List of replication	Each activity required to perform the replication must have a
	activities	description, specifying the dependency and order of the
		activities for the replicating experimenter.
	4.1.2 Estimated workload	()
	4.1.3 General schedule	()
4.2 Study	4.2.1 Objectives	Description of the high-level attributes and goals examined by
conception		the experiment.
	4.2.2 Hypotheses and sub	()
	studies	
	4.2.3 Factors and response	()
	variables	
	4.2.4 Contextual variables	()
4.3 Experi-	4.3.1 Design alternatives	List of design alternatives for the experiment. The alternatives
mental design		can include experimental designs already used in earlier
		replications that are potentially valid bearing in mind the
		study conception. They should include a reference to the
		experimental theory used as a basis for the design.
	4.3.2 Guidelines for	()
	selecting the experimental	
	design	
	4.3.3 Vallaation of the	[]
1 1 Operation	4.4.1 Instructions for	Stong pagaggary for proparing the material for the sessions
4.4 Operation	4.4.1 Instructions joi proparing material	This may involve printing out forms or setting up a laboratory
	preparing material	anyironment. They must specify what stops the realizating
		experimenter can take to improve the management of the
		material and assure the correct application of the experimental
		design They should include a checklist of deliverables that
		may vary depending on the treatment.
	4.4.2 Operatina material	()
	4.4.3 Instructions for	
	running sessions	

4.5 Analysis	4.5.1 Data collection	This section specifies the data collection method and any changes necessary to output the response variables. In some cases, several measurements may have to be taken to measure a response variable. All units and transformations must be specified. The final data are collected using standard templates with data fields for each factor and response variable. A spreadsheet template or similar should be included to make the replications easier to compare.
	4.5.2 Analysis methods	()
	4.5.3 Results interpretation	()

To use the LP template, we suggest that researchers start with simple instantiations of the core modules. The experiment module aims to capture the essential information needed to replicate an experiment. The other core modules —theory, training, evolution— should be used when needed to capture additional knowledge or when the experiment module contains a lot of information. These modules could be built after the initial effort to understand the baseline experiment. The training module is designed to include all the materials used in training. We have found that, in academic settings, experiments are usually conducted as part of a course, and it is common for educational materials (for both students and teachers) to add value, providing context for the replicating researcher.

5.3 Packaging checklist

The proposed LP structure is the groundwork for deploying other improvements to content and writing style. The incidents identified in the diagnosis stage show that many of the problems in the use of the LPs are due to information that is missing or not detailed enough for replication. The writing style is crucial for information comprehensibility and applicability. LPs should be written as *step-by-step instructions* focusing on activities that the replicating experimenter has to perform. On this ground, we drew up a number of documents detailing the LP structure and further specifying packaging criteria.

Apart from the LP template document, we provide a packaging checklist to be used by the researcher instantiating a LP for a specific experiment. The packaging checklist summarizes all the proposed improvements to the structure, content and writing style. The checklist is divided into four different key SE experiment packaging categories:

- Instructions for the replicating experimenter
- Operational material
- Experimental research process support
- Structural and usability improvements

Table 5 shows examples of checklist items for each category. The specific components of each improvement are added to the packaging checklist to assure that the researcher can easily check the instantiated LP. Some of the improvements are already addressed by the modularity of the LP template or in specific LP template sections. However, the packaging checklist could be used to check and improve the first draft of the LP. The packaging

checklist document with the components of each proposed improvement is included in the web appendix.

Category	Proposed	Components	LP template
	improvement		sections
Instructions for replicating experimenter	Replication plan (RP)	 List of activities and dependencies Estimation of times and resources by activity 	4.1 Planning
		Basic replication schedule	
	()		
Operational material	Sessions with time limit (ST)	 Maximum session time for subjects Short time limit (sessions lasting two or three hours) 	4.4.3 Instruc- tions for running sessions
	()		
Experimental research process support	Replication report (RR)	 Replication template: identification, characterization, results and lessons learned Modules added to the LP for each replication 	6.n.1 Descrip- tion of the replication
	()		
Structural and usability improvements	Navigation and search (NS)	 Conventional structures (index, table of contents, sections) Hyperlinks External references management Search engine 	(*) Transversal to the whole LP
	()		

Table 5: Excerpt from the packaging checklist.

The following discussion of the proposed improvements included in the different sections of the checklist serves to illustrate how to use the LP template and packaging checklist. For example, one of the identified experimental incidents was that experiment replications are very time-consuming and that the total workload is hard to estimate. On this ground, we suggest that the LP should contain a **replication plan**. This component is detailed in section *4.1 Planning* of the LP template.

Another example of an identified incident concerned experimental objects and session scheduling: replicating experimenters had trouble replicating the experiment within courses with time constraints. On this ground, we suggest that the operational material of the LP should include, whenever possible, other experimental objects (programs, models, test cases, etc.) to be used in **sessions with time limits** that can be adapted to the courses. This component is detailed in section *4.4.3 Instructions for running the sessions* in the LP template.

In a further example, the original DefDetect experiment LP contained no guidelines on how the replicating experimenter should report the replication. As a result, incidents such as very late replication reporting in wide-ranging formats were observed. This is an obstacle to the advancement of knowledge. On this ground, one of the improvements that we propose is a guideline for **replication reporting**. This component is detailed in section *6.n.1 Description of the replication* (*n* will be the number of replication study) of the LP template.

Regarding the LP format, some replicating experimenters drew attention to the fact that a lot of the material is hard to use. In view of the state of the art with regard to documentation, we suggest a minor improvement whereby the document should include **navigation and search facilities**. This proposed improvement is implemented in the structure itself and not in a particular section of the LP template.

6 Generating LPs for experiments

We arranged the experimental materials that the group of researchers had put together according to the proposed LP. This process relies on access to the knowledge on each experiment, including tacit knowledge that may not be openly documented. Even though the diagnosis stage improved our direct knowledge of the experiment enormously, we had to examine the available materials more thoroughly and meet with the experimenters in order to instantiate a LP.

In the case of experiments with which the person who built the LP was unfamiliar, this turned out to be a very demanding process. This makes sense since the researcher preparing a LP is expected to and should have been involved in running the experiment so it should not be considered a threat to our proposal. Two LPs were instantiated in Iteration 4. For the experiment on team and personality influence in development (TeamPersInf experiment), the researcher responsible for instantiating the LP was unfamiliar with the experiment. For the experiment on unit testing techniques (UnitTest experiment), the researcher responsible for generating the LP was unfamiliar with the LP structure proposal but had been involved in the experiment.

To help researchers (other than the authors of the proposal) apply the LP instantiation to different experiments, we generated a LP **instantiation procedure**. The aim of the procedure is to improve instantiation by providing support for processing the available materials (that may or may not be part of a pre-existing LP), as well as eliciting any tacit knowledge that needs to be specified. Figure 2 outlines the LP instantiation procedure.

The main sources for instantiating a LP are existing documents related to the experiment and tacit knowledge. The instantiation procedure relies on two tools that are part of the LP proposal: a LP template and a packaging checklist. After a number of intermediate results (restructured documents, list of missing components and new components), the end product is a LP instantiated for a specific experiment.

The first activity of the instantiation procedure involves **gathering existing documents** and material about an experiment. The LP template is an essential tool in this activity, since it details every section that should be included in a complete LP. For a person less experienced in experimentation, the LP template provides support in two ways. First, the explanations included for each section of the LP template inform about experimentation concepts. Second, the section structure guides in the identification of material about the experiment that should be gathered if available.

The LP template is used in the **completeness check** activity to generate a list of missing components. Using this list as a reference, the **generation of new components** activity involves interacting with researchers who may have the required knowledge and producing new components for the LP. Like any knowledge gathering procedure, the activities do not necessarily occur sequentially. However, by first analysing existing documents and checking the completeness against the proposed LP template, it is easier to identify the tacit knowledge to be gathered.

The packaging checklist is an additional tool for verifying the outcome. In the **review of changes activity**, the restructured documents and the generated components are checked against the packaging checklist. Many of the concepts included in the packaging checklist are transversal to the LP and not related to a specific section. For example, the writing style and the document format apply to the whole LP. The goal of the review is to improve the generated LP, not only aggregating existing documents and filling missing sections. As a result, researchers may be required to redo previous activities of the LP instantiation procedure.



Figure 2: LP instantiation procedure.

The experiment used as a case study for identifying the experimental incidents (Def-Detect) was the first family of experiments for which the proposed LP was instantiated. We (the authors of this article) are very familiar with this experiment. However, a new LP was instantiated in order to adapt the materials to the proposed LP designed in response to incidents detected in several previous replications. The instantiated LP was used in replications by researchers not familiar with the experiment. The second instantiated LP is a quasi-experiment on team and personality influence in development (TeamPersInf experiment) [50]. Random assignment is not possible in quasi-experiments. This experiment has several features that make it interesting for validating the proposed LP. Firstly, it is a complex quasi-experimental study, which includes a great many measured and analysed variables, measuring instruments, and instructions for analysing and interpreting results. The theory behind the study is backed by both psychology and SE. On the other hand, this is an advanced piece of research which has been internally replicated on a number of occasions, and has been reported in several published articles and a PhD thesis. It can therefore be considered as a mature piece of research providing all the knowledge required to elaborate a LP designed to externally replicate the study [57,58].

The third LP instantiation was conducted as part of an experiment on software unit testing techniques (UnitTest experiment) [59]. While this experiment studies testing techniques like the experiment that we used to identify incidents in replications, it is part of a different family of experiments conducted by another research group. The instantiation of this experiment has two goals: validate the proposed LP and put together a LP for use in future replications. From the viewpoint of the experiment, we took advantage of the validation to organize the information on an experiment and make it available for use in coming replications. The results of this instantiation with regard to both objectives were reported in [60]. In this case, the LP was instantiated by a researcher that did not author the LP proposal but was involved in running the experiment.

For the LP generation, we used a digital document that contains all the instantiated LP modules. On top of this, the operational material and the datasets from the replications are saved in a directory structure. One possible improvement in this respect would be to use web standards and a collaborative software infrastructure associated with the LP. At this stage of the research, we focused on the issue of structure and content irrespective of the implementing technology and used just a digital document and materials directories. Posting information about an experiment on the web makes it accessible to a wider audience. Unless it has the right structure and content for replication, however, its being on the web will not improve matters any, that is, as we have repeatedly found with published LPs like [16], the web does not solve the problem of which information a LP should contain.

7 Evaluation of the LP proposal

In the evaluation stage, we validated our LP proposal. On one hand, we checked that the LPs instantiated according to the proposal were useful for the research groups involved. On the other hand, we collected evidence about how the proposed LP supports conducting effective replications of experiments and reduces the number of incidents in the replications. Our evaluation observed the use of the proposed LP structure in different experiments and in different replications. The collected evidence shows the initial validity of the proposed LP and how it can be iteratively improved.

Our proposal was evaluated by dividing the overall objective of an LP into more specific aspects. We used two complementary approaches to evaluate the proposal: LP instantiation for different experiments and use of the instantiated LPs in replications. On one hand,

by instantiating the LP proposal, we check whether the proposal is capable of generating a LP for different families of experiments, that is, that the LP structure that we propose is **feasible** and **flexible**. On the other hand, we evaluate replications of experiments using LPs instantiated according to our proposal. By using LPs in replications, we can analyse how much support they provide for the experimental process activities, as well as how the replicating experimenter rates the quality of the document, that is, that the LP that we propose is **complete** and **usable**. These replications were evaluated according to the same procedure as in the diagnosis stage in order to compare the impact of using our proposal in experimental incidents), we can evaluate the **efficacy** and **efficiency** of our proposal, as well as its impact on the **satisfaction** of the replicating experimenter.

7.1 Evaluation of our LP proposal

The specific aim of the first evaluation activity is to check the **feasibility** of our LP proposal. We understand that the proposal is feasible if it can be instantiated as a LP for a specific experiment with a workload and outcome that is acceptable for experimenters. In the first iteration, we instantiated the DefDetect experiment that our research group is working on and wants to improve.

The **feasibility** criterion is divided into three more specific attributes: instantiation, instantiation workload and generated LP acceptance. We measured the workload attribute in person-hours. For the instantiation and acceptance attributes, we used a binary scale to simplify the evaluation process: either the generated LP was instantiated and accepted by the experimenters or not. Other aspects of the generated LP, as the document quality and ease of use are evaluated by other attributes. The result of this evaluation activity was the successful generation of a LP instance for the DefDetect experiment in return for a reasonable amount of effort on the part of the experimenters. On this ground, we consider our LP proposal to be workable.

We then instantiated our LP proposal for the TeamPersInf and UnitTest experiments. These instances have the potential of corroborating the feasibility evaluation of the proposal. However, the key goal of instantiating these LPs is to check that the proposal is **flexible** enough to adapt to different experiments. Many things may vary from one experiment to another and from one context to another: the characteristics of the experimental study, the knowledge available for instantiation and the context in which the experiment is run, among others. In order to evaluate flexibility, we thoroughly examine the differences between the generated LPs to check whether they account for the specified characteristics of each family of experiments. In particular, we looked at whether sections had to be added or removed in order to instantiate a LP for each experiment.

The flexibility of the proposed LP is not an absolute term. We found that a LP instantiating our proposal can be generated for three families of experiments, whence we inferred that the proposal may be applicable in other cases. As LPs are generated (by instantiating our proposal) for more experiments, the flexibility of our proposal is further corroborated. If the LP is instantiated for other families of experiments, our proposal may require adjustment. Each experiment is likely to have particular characteristics and require a specific treatment not accounted for in the original proposal. Such changes are unavoidable, as a

high-level solution cannot be expected to account for each and every case. Any such changes that occur will have to be documented and analysed. Whenever appropriate, they may be integrated into the basic LP proposal.

The TeamPersInf and UnitTest experiments differ from DefDetect experiment in different respects. The TeamPersInf experiment deals with another process area within SE. The study method is a quasi-experiment, which is quite far removed from the original controlled experiment. As the UnitTest experiment this is again a controlled experiment on software testing techniques, the differences have less to do with the topic of study or the experimental design than with the person conducting the instantiation and the knowledge available about the experiment.

In both instantiations, we observed that the instantiation effort is within an acceptable range (from 75 to 85 hours). This considers the amount of effort required to adapt the LP structure and generate new documentation for each experiment. This observation validates the feasibility issue already confirmed by the first instantiation. We also found that there is not a big difference in the instantiation effort for a person unfamiliar with the proposed LP.

7.2 Evaluation of the use of the LP in replications of experiments

The aspects evaluated in this activity focus on the use of the LP during replication. In particular, we evaluate the **completeness** and **usability** of the documentation based on our proposed LP. We examine whether it supports all the activities of the experimental process and how usable it is from the viewpoint of the researcher who uses it to replicate an experiment. On the other hand, we evaluate the **efficacy** and **efficiency** of the replication, and the subjective experimenter **satisfaction**. It is hard to evaluate these aspects through a simple static analysis of the instantiated LPs. Ideally, the use of the LP should be observed within the framework of an experiment replication.

In order to evaluate the replications, we applied the same empirical evaluation procedure to the replications as we used in the diagnosis stage of this research. By strictly adhering to the same evaluation procedure, the result will be comparable to the replications evaluated in the previous iterations. Specifically, we can observe whether the LP prevents or mitigates the experimental incidents identified in replications where our LP was not used. This does not mean that we have found the ultimate solution to the LP problem, but it does at least raise the confidence that we have in our proposal improving upon the current quality of the information used to replicate experiments.

The evaluated replications used LPs instantiated according to our proposal for the Def-Detect experiment, the same experiment used in the replications of the diagnosis stage of Iterations 1 and 2. The exact objective of using the same experiment is to improve the LP for the research group. On the other hand, by using an improved LP in later replications of the same experiment, we were able to compare the impact of LP use directly.

The evaluation included two replications, each of which was performed in a different context and by a different replicating experimenter. The replicating experimenters involved in each replication are unfamiliar with the details of the experiment. This is a necessary condition of the evaluation, as, if they were acquainted with the experiment, we would not be able to evaluate the impact of the LP as a knowledge transmission instrument. In both cases, we made contact with the participating researchers, thanks to which we were able to evaluate the replication at close quarters, albeit without intervening in its execution. Running two replications will increase the validity and reliability of the results.

The experimenter using the LP in each replication rated document usability directly using a five-value Likert scale, where 5 is best and 1 is worst. On this ground, although similar LPs were used, the rating varies considerably across replications. However, we can get an idea of document usability as perceived by the participating researcher by taking the mean rating scores.

Table 6 shows the values for the usability attributes and components of all the evaluated replications. The columns shaded grey refer to the replications using LPs structured according to this proposal. For the purposes of comparing the numerical usability values, the table includes two columns listing the means before (PRE) and after (POST) the application of the proposed LP. The highest scores for all the usability attributes and components were recorded when the LP proposal was used. This applies to differing extents depending on the different attributes. We can, however, conclude that the usability of the document improves if the proposed LP is used.

Attribute	Component	UPM 2000	UPM 2004	UdS 2005	ORT 2005	UPV 2006	ORT 2009	UPM 2011	Mean PRE	Mean POST
Ease of application	Task orienta- tion	2	1	5	2	4	5	4	2.8	4.5
	Accuracy	4	3	3	4	4	5	4	3.6	4.5
	Complete- ness	2	1	3	2	3	5	4	2.2	4.5
Ease of understand-	Clarity	3	1	5	3	4	4	5	3.2	4.5
ing	Concretion	3	1	4	2	5	4	5	3	4.5
	Style	3	1	5	4	4	4	4	3.4	4
Ease of search	Organization	3	1	5	3	5	4	5	3.4	4.5
	Retrievabil- ity	3	3	5	2	4	4	3	3.4	3.5
	Visual effectiveness	1.5	1	3.5	2	2	4	4	2	4
Values assigned	d by the experim	enter usi	ng the LP	(5-point	Likert sca	ale where	= 1=worst,	5=best)		
Columns shade	d grey refer to tl	ne replica	tions usir	ng LPs str	uctured a	according	to this pr	oposal		

Table 6: Summary of LP usability evaluation.

Table 7 shows the values for the completeness attributes and components of all the evaluated replications. The columns shaded grey refer to the replications using LPs structured according to this proposal. The comparative study of completeness illustrates the evolution of the LP. The replications using LPs structured according to the proposal provide more coverage of experimental process activities. The scope and level of instruction of these LPs is better. Albeit to a lesser extent, we found that experimental designs are more easily adaptable using the LP proposal. Using the structured LPs, a replication report can be drawn up faster and then integrated into the LP itself. We can conclude that the LP structured according to the proposal is more thorough than the non-structured variants.

Attribute / Compo-	UPM	UPM	UdS	ORT	UPV	ORT	UPM
nent	2000	2004	2005	2005	2006	2009	2011
Scope	Opera-	Training	Training	Desiin	Design	Complete	Complete
Values: Training,	tional	Design	Design	Opera-	Opera-		
Design, Operational,		Opera-	Opera-	tional	tional		
Complete		tional	tional				
Level of instruction	Materials	Basic	Detailed	Basic	Detailed	Detailed -	Detailed -
Values: Materials only,	only					Grounded	Grounded
Basic, Detailed,							
Detailed – Grounded							
Adaptability	Un-	Un-	Partially	Un-	Partially	Adapt.	Partial
Values: Unchangeable,	change-	change-	adapt.	change-	adapt.		adapt.
Partially adaptable,	able	able		able			
Adaptable		-					
Scalability /	Late	Late	No	Late	No	Yes	Yes
Replication reported							
Values: Yes, No, Late							
Scalability / Aggre-	Partial	Yes	No	No	Partial	Partial	Partial
gated results							
Values: Yes, No, Partial			-				
Version control	No	Current	Current	Current	Current	Current	Retrieva-
Values: No, Current		version	version	version	version	version	ble
version only, Log,		only + log	only	only	only	only	versions
Retrievable versions.							
More than one element	could be assig	gned for the S	Scope attribu	te (Complete	includes all e	elements).	
For the other attributes,	the possible	values are or	dered worst	to best.			
Columns shaded grey re	fer to the rep	lications usir	ng LPs structi	ured accordir	ng to this pro	posal	

 Table 7: Summary of LP completeness evaluation.

The main objective of an LP is to answer replicating researchers' questions in order to reproduce the intended environment for the experiment. An improved LP should mitigate experimental incidents, lowering the mean error severity. The efficacy of the LP as a knowledge transfer instrument was evaluated by means of three qualitative attributes (as in the diagnosis stage): question answering, environment reproduction and mean severity of experimental errors. To compare each replication, we used a three-level scale to measure each attribute. Error severity is regarded as high if an experimental incident prevents the integration of the experimental results into the experimental body of knowledge of the respective research project. Error severity is required for results integration. A low mean error severity occurs when the incidents do not much alter the course of the replication and do not affect the results. We used the same replication evaluation instruments and procedure to determine these qualitative attributes in both the diagnosis and the evaluation stages (semi-structured interviews with experimenters, document analysis and additional questions).

Table 8 shows a summary rating these attributes across seven evaluated replications. The two replications that use the LP instantiated according to our proposal are highlighted in grey. The two replications conducted with the LP based on our proposal yield better

results than the previous replications. In practical terms, this means that we identified fewer incidents overall and the severity of the incidents was lower.

Iteration	Iteration 1	Iteration 2	Iteration 2 Iteration 3						
Research stage	Diagnosis	Diagnosis				Evaluation	1		
LP version	Without LP	Previous v	version			Our propo	sal		
	UPM 2000	UPM	UdS	ORT	UPV	ORT	UPM		
		2004	2005	2005	2006	2009	2011		
Question answer-	Low	Complete	Medium	Low	Medium	High	High		
ing									
Environment	N/A	Complete	Low	Low	Medium	Complete	Complete		
reproduction									
Mean error	Medium	Slight	Serious	Medium	Medium	Slight	Slight		
severity									
Columns shaded gre	Columns shaded grey refer to the replications using LPs structured according to this proposal.								

Table 8: Summary of LP efficacy evaluation.

During the replications, we also found that the usability and completeness of the document used as the LP was perceived to be more positive than in the replications that do not use the proposal. These issues were evaluated taking into account the opinion of the researcher involved in the replication (none of whom were involved in putting together the proposed LP) and the ultimate analysis of the document content.

The efficiency was measured by asking researchers replicating the experiments using the improved LP to log her effort in person-hours. Reliable estimations of the effort needed for replicating the experiment before the LP proposal were not available. However, we tried to reconstruct the replication process in the interviews with the replicating experimenters. Although we are not able to provide a rigorous evaluation of the efficiency of the proposed LP, we did observe that the effort was lower to the estimate effort for previous replications.

Each evaluation creates a detailed analysis of incidents and LP use. These experiments and replications checked that our proposal meets the following conditions:

- 1. The LP can be instantiated according to our proposal for different SE experiments (feasibility and flexibility).
- 2. The quality of the instantiated LPs is satisfactory (completeness and usability).
- 3. The use of the LP improves the experimental process (efficacy, efficiency and replicator satisfaction).

As limitations of this proposal, we should note that it only provides partial support for aggregation studies. It is possible to use the template to include secondary studies, but this has not been validated in either of the two replication studies. The version control proposal (including multiple versions of the experiment in the same LP) was used in one of the replications. However, we found that these multiple retrievable versions were not used to plan the actual replication. Experiment change management is a much more complex issue that we expected. Therefore, another separate research project was initiated to further explore this issue.

We admit that less experienced researchers might find it challenging to apply the LP structure proposal. However, the researchers responsible for the evaluated replications were graduate students without much experimental experience. Overall, they successfully instantiated the LP and replicated the experiment without direct intervention.

8 Generalization of our LP proposal

Experiment replication is a demanding process. On top of this, experimenters and information on an experiment are not easily accessible when conducting research like this. Three experiments were considered throughout this research. We looked for experiments that differed significantly with respect to the procedures and materials used. Even so, not many changes had to be made to the LP. Taking into account just how limited access to experiments is, we consider that the above three experiments constitute a big enough sample for us to be able to claim that our LP proposal is sufficiently generalizable across experiments.

Another component of the replication, apart from the experiment itself, is the replicating researcher. This is a key component of our research, as the researcher is precisely the knowledge transfer target. Subjectivity comes into play when each researcher interacts with the LP. To address this concern, we engaged seven different replicating experimenters in the replications analysed in our research. One necessary premise for evaluating the impact of the LP is that researchers should be unfamiliar with the experiment at the time of conducting the replication in question. This means that we can study the most troublesome (but quite common) instance of information transfer in a replication: when the replicator is completely unfamiliar with the experiment. Besides, the seven researchers have differing levels of experience in experimentation: some are experts, whereas this is their first contact with a SE experiment for others. In all cases, we gathered information about other mechanisms of communication apart from the LP that were used in each replication. Although we cannot control all human factors and means of communication having an impact on a replication, we believe that we have mitigated this threat by accounting for a diverse set of replicating experimenters and replications and, as far as possible, taking note of the knowledge transfer instruments. Therefore, we take the view that, having studied seven different replicating experimenters in seven replications, the generalizability of our proposal with respect to experimenters is acceptable.

The findings with respect to the LP structure and content are limited by the number and diversity of the experiments packaged using our proposal, as well as by the researchers participating in the replications studied. Across the research iterations, three different experiments were packaged and seven replications conducted by seven different researchers at five different sites were evaluated. Thus, we are confident that our LP proposal is generalizable enough to be able to be applied to other SE experiments.

Regarding the cultural environment of this research project, all the evaluated replications were conducted in Spanish. The LP proposal and its instantiation for specific experiments were also written in Spanish. In view of these constraints, the transferability of the proposed solution to other environments requires further consideration. We do not think that the LP proposal has any cultural or language-specific terms. However, the translation of

the LP template and packaging checklist should be evaluated in practice by means of a pilot replication study.

One validity threat to our proposal is our connection with the authors of the experiments and evaluated researchers. Two of the authors of this paper were responsible for two replications (Vegas in UPM 2000 and Solari in ORT 2005). The person that collected and analysed the data for evaluating all the replications (Solari) was also a member of the team that put together the LP proposal. There is, therefore, a risk of self-confirmation in any case where a researcher tests his or her own method or solution. To mitigate this threat, we defined a replication evaluation procedure beforehand. This procedure specifies activities, intermediate products and coded value levels. The results can be traced back to and reproduced from the defined information sources. For the open coding, we linked all the codes to literal citations from interviews and documents applied in the replication. The results of each replication evaluation were validated by the principal investigator responsible for the replication or the experiment. At the validation stage of the proposed LP, we took care not to intervene in the replication so as not to influence knowledge transfer in any way. However, this threat cannot be completely eradicated because the replicating experimenter is aware that the replication and particularly the LP are being evaluated.

We do not claim that the resulting proposal is universally applicable or succeeds in fully supporting replication. As opposed to pure scientific research, the aim of technological research is to find solutions that improve upon the current situation. The evaluation that we conducted shows that the proposal is acceptable and to some extent reduces the amount of effort currently required to replicate experiments in SE. We do not claim that this proposal is the only or even the best possible LP structure. It is a structure that has been satisfactorily validated and can thus be recommended for use in practice.

The further validation and evolution of this proposal requires instantiations of experiments from different SE domains and more replications using structured LPs. We would like to call on collaborators to add more experiments. All the elements of the proposal are publicly available from our website¹⁴. We would be happy to provide support for other researchers using the LP template and packaging checklist to instantiate their own experiments. We are also willing to instantiate a LP for experimenters who send in experiment materials. Experimenters would only be required to be available for tacit knowledge elicitation. We would like to receive any feedback on the instantiation process and, if possible, evaluate the replications using the structured LP. Each and every experiment and replication context will improve the proposal and foster collaboration among experimenters to replicate experiments.

9 Further evolution of the LP proposal

This version of the LP proposal is implemented in a set of files (Introduction, LP template document, Packaging checklist document). We chose this format for simplicity: the single file documents can be downloaded and edited using a popular word processor without requiring specific knowledge. However, we think that future versions of the LP template

¹⁴ http://www.grise.upm.es/LaboratoryPackage

should be based on Web standards. The hypertextual support provided by Web standards would be better suited to the modular LP structure. A standardized format would improve the general portability of the LP. However, this change should be evaluated carefully against the availability and ease of use of editing tools.

Another possible way of evolving the LP proposal would be its implementation within an online collaboration platform. The platform would enable the creation and publication of LPs for different experiments that could be easily transferred and used. It would also provide communication among researchers in parallel to LP use. All the participants in an experimentation project would be able to view questions and answers. The ultimate aim of this proposal is to foster replications that can be integrated into a SE body of knowledge. An online collaboration platform would speed up that process. On the other hand, such a platform would raise user authentication, security and other issues that can be hard to implement.

A LP is used in scenarios where there are different levels of communication among researchers. We have found that it is harder to understand the LP and conduct the replication if there is no communication between the experimenters running the baseline experiment and the replication. The results of a replication where there has been no communication among experimenters are harder to interpret and integrate into the broader experimental body of knowledge of the respective line of research. We believe that this LP proposal could yield better results if it is used in combination with communication mechanisms while performing the replication. We have found that an experimental design validation meeting and a results interpretation meeting at least are necessary to produce useful replications. The LP proposal could be further developed to complement the interresearcher communication mechanisms. To do this, we suggest that the communications among researchers should be included in the LP history and the useful experiences should be assessed in order to evolve the communication section.

We decided to focus our LP proposal on controlled experiments. Controlled experiments are a common thorough approach within the experimental scientific method. Controlled experimentation requires the precise definition of techniques, measurement and context. In some cases, replications depend on the researcher conducting the training, assignment, monitoring or data analysis according to predefined criteria. During the evaluation phase, we instantiated the LP proposal for two controlled experiments and one quasi-experiment. We found that the proposal is flexible with respect to the experimental approach. However, other empirical studies may require different approaches. The LP would have to be instantiated for case studies and other observational studies with different degrees of intervention in order to further evolve the proposal. We believe that most of the modules in the LP template and the packaging checklist would be useful for a wide range of scenarios, but the proposal would have to be evolved based upon a thorough analysis.

10 Conclusion

Experimentation is a key strategy for maturing knowledge in any area of science and technology. SE needs experimentation in order to provide reliable evidence about techniques, methods and tools. However, an individual experiment does not provide unques-

tionable and irrefutable results. Experimental research is an incremental process at the heart of which is the replication of experiments. Experiment replication cannot be done without the transfer of knowledge among researchers, and this is one of the major challenges now facing empirical research in SE.

If replications are to be less costly and complex, special-purpose instruments are required to carry out experimentation in SE. LPs are one of the principal instruments for supporting knowledge transfer and running replications. We have put together and evaluated a LP proposal for SE experiments. The purpose of this research is to demonstrate that the replications run using the experiment LPs instantiated according to our proposal are better than replications using unstructured or ad hoc LPs which are what our community tends to use now.

We applied a set of defined stages iteratively to generate and evaluate our proposed LP. Thanks to this approach, we were able generate LPs for different experiments, as well as a LP structure and content proposal for SE experiments. We have generated LPs for different experiments and evaluated replications using the instantiated LPs.

Throughout this research, we evaluated seven replications. These replications accounted for three different families of experiments and five replication sites. In all cases, the replications were conducted by researchers who were unfamiliar with the replicated experiment. In each case, we evaluated different aspects of our proposed LP. Our proposal proved to be good enough to iteratively improve these three families of experiments.

11 References

- [1] C. Wohlin, P. Runeson, M. Höst, M.C. Ohlsson, B. Regnell, A. Wesslén, Experimentation in Software Engineering, Springer, 2012.
- [2] V.R. Basili, R.W. Selby, D.H. Hutchens, Experimentation in Software Engineering, IEEE Trans. Softw. Eng. 12 (1986) 733–743.
- [3] F. Shull, J.C. Carver, S. Vegas, N. Juristo, The role of replications in Empirical Software Engineering, Empir. Softw. Eng. 13 (2008) 211–218. doi:10.1007/s10664-008-9060-1.
- [4] O.S. Gómez, N. Juristo, S. Vegas, Understanding replication of experiments in software engineering: A classification, Inf. Softw. Technol. 56 (2014) 1033–1048. doi:10.1016/j.infsof.2014.04.004.
- [5] J. Miller, Triangulation as a basis for knowledge discovery in software engineering, Empir. Softw. Eng. 13 (2008) 223–228. doi:10.1007/s10664-008-9063-y.
- [6] J.C. Carver, N. Juristo, M.T. Baldassarre, S. Vegas, Replications of software engineering experiments, Empir. Softw. Eng. 19 (2014) 267–276. doi:10.1007/s10664-013-9290-8.
- [7] A. Brooks, J. Daly, J. Miller, M. Roper, M. Wood, Replication of experimental results in software engineering, ISERN Report, 1996.
- [8] V.R. Basili, F. Shull, F. Lanubile, Building knowledge through families of experiments, IEEE Trans. Softw. Eng. 25 (1999) 456–473. doi:10.1109/32.799939.

- [9] S. Vegas, N. Juristo, A.M. Moreno, M. Solari, P. Letelier, Analysis of the influence of communication between researchers on experiment replication, in: Int. Symp. Empir. Softw. Eng. (ISESE 2006), ACM Press, New York, New York, USA, 2006: p. 28. doi:10.1145/1159733.1159741.
- [10] N. Juristo, S. Vegas, M. Solari, S. Abrahao, I. Ramos, A Process for Managing Interaction between Experimenters to Get Useful Similar Replications, Inf. Softw. Technol. 55 (2013) 215–225.
- [11] R. Moonesinghe, M.J. Khoury, A.C.J.W. Janssens, Most published research findings are false-but a little replication goes a long way., PLoS Med. 4 (2007) e28. doi:10.1371/journal.pmed.0040028.
- [12] S. Schmidt, Shall we really do it again? The powerful concept of replication is neglected in the social sciences, Rev. Gen. Psychol. 13 (2009).
- [13] H. Collins, Changing Order: Replication and Induction in Scientific Practice, University Of Chicago Press, 1992.
- [14] F. Godlee, Publishing study protocols: making them visible will improve registration, reporting and recruitment, BMC News Views. 2 (2001) 4.
- [15] J. Giles, The trouble with replication., Nature. 442 (2006) 344–7. doi:10.1038/442344a.
- [16] F. Shull, M.G. Mendonça, V.R. Basili, J.C. Carver, J.C. Maldonado, S. Fabbri, G.H. Travassos, M.C. Ferreira, Knowledge-Sharing Issues in Experimental Software Engineering, Empir. Softw. Eng. 9 (2004) 111–137. doi:10.1023/B:EMSE.0000013516.80487.33.
- [17] F.Q.B. da Silva, M. Suassuna, A.C.C. França, A.M. Grubb, T.B. Gouveia, C.V.F. Monteiro, I.E. dos Santos, Replication of empirical studies in software engineering research: a systematic mapping study, Empir. Softw. Eng. (2014) 501–557. doi:10.1007/s10664-012-9227-7.
- [18] M. Solari, S. Vegas, Classifying and analysing replication packages for software engineering experimentation, in: Work. Empir. Softw. Eng. 7th Int. Conf. Prod. Focus. Softw. Process Improv. Amsterdam, Holl., 2006. http://grise.upm.es/docs/WSESE_Classifying_and_analysing.pdf (accessed October 14, 2011).
- [19] ClinicalTrials.gov, Understanding Clinical Trials, Natl. Inst. Heal. (2015).
- [20] I. Chalmers, D.G. Altman, How can medical journals help prevent poor medical research? Some opportunities presented by electronic publishing, Lancet. 353 (1999) 490–493.
- [21] K.F. Schulz, D.G. Altman, D. Moher, CONSORT 2010 statement: updated guidelines for reporting parallel group randomized trials, Ann. Intern. Med. 152 (2010) 726– 732.
- [22] A.-W. Chan, Bias, spin, and misreporting: time for full access to trial protocols and results, PLoS Med. 5 (2008) e230.
- [23] J.M. Tetzlaff, A.-W. Chan, J. Kitchen, M. Sampson, A.C. Tricco, D. Moher, Guidelines for randomized clinical trial protocol content: a systematic review, Syst Rev. 1 (2012) 43.
- [24] A.-W. Chan, J.M. Tetzlaff, D.G. Altman, A. Laupacis, P.C. Gøtzsche, K. Krleža-Jerić, A. Hróbjartsson, H. Mann, K. Dickersin, J.A. Berlin, C.J. Doré, W.R. Parulekar, W.S.M.

Summerskill, T. Groves, K.F. Schulz, H.C. Sox, F.W. Rockhold, D. Rennie, D. Moher, SPIRIT 2013 Statement: Defining Standard Protocol Items for Clinical Trials, Ann. Intern. Med. 158 (2013) 200–207. http://dx.doi.org/10.7326/0003-4819-158-3-201302050-00583.

- [25] R.M. Lindsay, A.S.C. Ehrenberg, The Design of Replicated Studies, Am. Stat. 47 (1993).
 http://www.tandfonline.com/doi/abs/10.1080/00031305.1993.10475983#.VLT8 uIrF_UI (accessed January 13, 2015).
- [26] V.R. Basili, S. Green, O. Laitenberger, F. Lanubile, F. Shull, S. Sorumgard, M. V. Zelkowitz, S. Sorumgaard, Packaging Researcher Experience to Assist Replication of Experiments, 1996.
- [27] F. Shull, V.R. Basili, J.C. Carver, J.C. Maldonado, G.H. Travassos, M.G. Mendonça, S. Fabbri, Replicating Software Engineering Experiments: Addressing the Tacit Knowledge Problem, Int. Symp. Empir. Softw. Eng. (2002). http://portal.acm.org/citation.cfm?id=857197.857885.
- [28] A. Jedlitschka, M. Ciolkowski, D. Pfahl, Reporting guidelines for controlled experiments in software engineering, in: F. Shull, J. Singer, D.I. Sjoberg (Eds.), Guid. to Adv. Empir. Softw. Eng., Springer, 2008.
- [29] J.C. Carver, Towards Reporting Guidelines for Experimental Replications: A Proposal, in: 1st Int. Work. Replication Empir. Softw. Eng. Res. [Held Dur. ICSE 2010], 2010.
- [30] P. Runeson, A. Stefik, A. Andrews, Variation factors in the design and analysis of replicated controlled experiments, Empir. Softw. Eng. 19 (2013) 1781–1808. doi:10.1007/s10664-013-9262-z.
- [31] M. Solari, Propuesta de Paquete de Laboratorio para Experimentos de Ingeniería de Software, Universidad Politécnica de Madrid, 2012. http://oa.upm.es/10171/.
- [32] M. Genero, J.A. Cruz-Lemus, D. Caivano, S. Abrahao, E. Insfran, A. Carsí, Does the use of stereotypes improve the comprehension of UML sequence diagrams?, Int. Symp. Empir. Softw. Eng. Meas. (2008) 300–302. doi:10.1145/1414004.1414059.
- [33] H. Do, S. Elbaum, G. Rothermel, Supporting Controlled Experimentation with Testing Techniques: An Infrastructure and its Potential Impact, Empir. Softw. Eng. 10 (2005) 405–435. doi:10.1007/s10664-005-3861-2.
- [34] P.G. Mian, G.H. Travassos, A.R.C. Rocha, A Computerized Infrastructure for Supporting Experimentation, Exp. Softw. Eng. Lat. Am. Work. (ESELAW 2005). (2005).
- [35] D.I.K. Sjøberg, I.C. Society, J.E. Hannay, O. Hansen, V.B. Kampenes, A. Karahasanovic, N.K. Liborg, A.C. Rekdal, A survey of controlled experiments in software engineering, IEEE Trans. Softw. Eng. 31 (2005) 733–753. doi:10.1109/TSE.2005.97.
- [36] E. Kamsties, C.M. Lott, An Empirical Evaluation of Three Defect-Detection Techniques, Eur. Softw. Eng. Conf. (1995).
- [37] V.R. Basili, S. Green, O. Laitenberger, F. Lanubile, F. Shull, S. Sørumgård, M. V. Zelkowitz, The Empirical Investigation of Perspective-Based-Reading, Empir. Softw. Eng. 1 (1996) 133–164.
- [38] T. Thelin, P. Runeson, C. Wohlin, An Experimental Comparison of Usage-Based and Checklist-Based Reading, IEEE Trans. Softw. Eng. 29 (2003) 687–704.

- [39] A. Dunsmore, Lab Package: Replication of Experiments concerning the Inspection Of Object-Oriented Code, 2002.
- [40] B. Du Bois, S. Demeyer, J. Verelst, T. Mens, M. Temmerman, Does God Class Decomposition Affect Comprehensibility?, in: IASTED Int. Conf. Softw. Eng., Innsbruck, Austria, 2006: pp. 346–355.
- [41] V.R. Basili, R.W. Selby, Comparing the effectiveness of software testing strategies, Department of Computer Science, University of Maryland - College Park, 1985.
- [42] M. Roper, M. Wood, J. Miller, An empirical evaluation of defect detection techniques, Inf. Softw. Technol. 39 (1997) 763–775.
- [43] N. Juristo, S. Vegas, Functional Testing, Structural Testing, and Code Reading: What Fault Type Do They Each Detect?, in: A.I. Wang (Ed.), Empir. Methods Stud. Softw. Eng. - Exp. from Eser., Springer, 2003: pp. 208–232.
- [44] N. Juristo, S. Vegas, M. Solari, S. Abrahao, I. Ramos, Comparing the Effectiveness of Equivalence Partitioning, Branch Testing and Code Reading by Stepwise Abstraction Applied by Subjects, in: Int. Conf. Softw. Testing, Verif. Valid. (ICST 2012), 2012. doi:10.1109/ICST.2012.45.
- [45] M. Solari, S. Matalonga, A Controlled Experiment to Explore Potentially Undetectable Defects for Testing Techniques, in: Int. Conf. Softw. Eng. Knowl. Eng. (SEKE 2014), Vancouver, Canada, 2014.
- [46] C. Apa, O. Dieste, E.G. Espinosa, E.R. Fonseca, Effectiveness for detecting faults within and outside the scope of testing techniques: an independent replication, Empir. Softw. Eng. 19 (2014) 378–417. doi:DOI 10.1007/s10664-013-9267-7.
- [47] O.S. Gómez, S. Vegas, N. Juristo, Impact of CS Programs on the Quality of Test Cases Generation: An Empirical Study, in: Softw. Eng. Educ. Train. Track (ICSE-SEET 2016), 2016.
- [48] M. Solari, Identifying Experimental Incidents in Software Engineering Replications, in: Int. Symp. Empir. Softw. Eng. Meas. (ESEM 2013), IEEE, 2013: pp. 213–222. doi:10.1109/ESEM.2013.26.
- [49] V.R. Basili, R.W. Selby, Comparing the Effectiveness of Software Testing Strategies, IEEE Trans. Softw. Eng. SE-13 (1987) 1278–1296.
- [50] S.T. Acuña, M.N. Gómez, N. Juristo, Towards understanding the relationship between team climate and software quality--a quasi-experimental study, Empir. Softw. Eng. 13 (2008) 401–434. doi:10.1007/s10664-008-9074-8.
- [51] D. Vallespir, S. Moreno, C. Bogado, J. Herbert, Towards a Framework to Compare Formal Experiments that Evaluate Testing Techniques, Res. Comput. Sci. (2009).
- [52] S. Kvale, InterViews: An Introduction to Qualitative Research Interviewing, Sage Publications, 1996.
- [53] N. Juristo, A.M. Moreno, Basics of Software Engineerig Experimentation, Kluwer, 2001.
- [54] IBM, Producing Quality Technical Information, Santa Teresa, CA, 1983.
- [55] A.J. Marlow, Quality Control for Technical Documentation, Moonprint, 2005.
- [56] B.J. Rosenberg, Technical writing for engineers and scientists, Addison-Wesley, Upper Saddle River, 2005.

- [57] S.T. Acuña, M.N. Gómez, J. de Lara, Empirical study of how personality, team processes and task characteristics relate to satisfaction and software quality, Int. Symp. Empir. Softw. Eng. Meas. (ESEM 2008). (2008) 291–293. doi:10.1145/1414004.1414056.
- [58] S.T. Acuña, M.N. Gómez, N. Juristo, How do personality, team processes and task characteristics relate to job satisfaction and software quality?, Inf. Softw. Technol. 51 (2009) 627–639. doi:10.1016/j.infsof.2008.08.006.
- [59] D. Vallespir, C. Apa, S. De León, R. Robaina, J. Herbert, Effectiveness of five verification techniques, in: XXVIII Int. Conf. Chil. Comput. Soc., 2009.
- [60] C. Apa, M. Solari, D. Vallespir, S. Vegas, Construcción de un Paquete de Laboratorio para un Experimento en Ingeniería de Software, in: XIV Ibero-American Conf. Softw. Eng. (CIbSE 2011), 2011.